\documentclass[11pt,reqno]{amsart}
\usepackage[utf8]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{517.956.3} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{35A01, 35A02, 35A09, 35L15, 35L71} %Here the author places classificators according to the AMS classification list.
\setcounter{page}{1}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{assertion}{Assertion}

\hypersetup{linkcolor=blue, filecolor=black, urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{99} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).

\bibitem{mech} J. Wang, J. Liu, G. Pan,
\href{https://doi.org/10.1103/PhysRevE.94.042218}{\it A nonlinear vibration control of a string using the method based on its time-varying length}, Machines, {\bf 10}:11 (2022), 981.

\bibitem{mech2} M. Devaud, T. Hocquet,
\href{https://hal.science/hal-01532358v2}{\it An essay on the vibrating string and its nonlinearities}, hal-01532358v2 (2017).

\bibitem{thermo} V.A. Sabelnikov, N.N. Petrova, A.N. Lipatnikov,
\href{https://doi.org/10.1103/PhysRevE.94.042218}{\it Analytical and numerical study of travelling waves using the Maxwell-Cattaneo relaxation model extended to reaction-advection-diffusion systems}, Phys. Rev. E, {\bf 94} (2016), 042218.

\bibitem{cau_0} S.N. Baranovskaya,
\href{https://www.mathnet.ru/php/archive.phtml?wshow=paper&jrnid=de&paperid=7523}{\it On classical solution of a mixed problem for a one-dimensional hyperbolic equation}, Differ. Uravn., {\bf 27}:6 (1991), 1071--1073.

\bibitem{cau_1} V.I. Korzyuk, I.I. Stolyarchuk,
\href{https://doi.org/10.1134/S0012266117010074}{\it Classical solution of the first mixed problem for second-order hyperbolic equation in curvilinear half-strip with variable coefficients}, Diff. Equat., {\bf 53} (2017), 74--85.

\bibitem{cau_2} W. Shi, K. Liu,
\href{https://doi.org/10.1016/j.aml.2018.05.009}{\it A new analytical formula for the wave equations with variable coefficients}, Applied Mathematics Letters, {\bf 84} (2018), 137--142.

\bibitem{cau_3} N. Nakagawa, Y. Shintaro,
\href{https://doi.org/10.11948/20220325}{\it Well-posedness of wave equation with a variable coefficient by method of characteristics}, Journal of Applied Analysis \& Computation, {\bf 13}:5 (2023), 2408--2415.

\bibitem{cau_4} P.-F. Yao,
\href{https://doi.org/10.1137/S0363012997331482}{\it On the observability inequalities for exact controllability of wave equations with variable coefficients}, SIAM Journal on Control and Optimization, {\bf 37}:5 (1999), 1568--1599.

\bibitem{cau_5} X. Fu, Z. Liao,
\href{https://doi.org/10.1137/21M1468231}{\it Observability estimate for the wave equation with Variable coefficients}, SIAM Journal on Control and Optimization, {\bf 60}:4 (2022), 2344--2372.

\bibitem{cau_6} J. Metcalfe, D. Tataru,
\href{https://doi.org/10.1007/s00208-011-0714-8}{\it Global parametrices and dispersive estimates for variable coefficient wave equations}, Mathematische Annalen, {\bf 353}:4 (2011), 1183--1237.

\bibitem{cau_7} G. Todorova, B. Yordanov,
\href{https://doi.org/10.1016/j.jde.2009.03.020}{\it Weighted $L^2$-estimates for dissipative wave equations with variable coefficients}, Journal of Differential Equations, {\bf 246}:12 (2009), 4497--4518.

\bibitem{cau_nl_1} V.I. Korzyuk, J.V. Rudzko,
\href{https://doi.org/10.29235/1561-8323-2023-67-1-14-19}{\it Classical solution of the initial-value problem for a one-dimensional quasilinear wave equation}, Doklady of the National Academy of Sciences of Belarus, {\bf 67}:1 (2023), 14--19.

\bibitem{cau_nl_2} V.I. Korzyuk, J.V. Rudzko,
\href{https://doi.org/10.29235/1561-8323-2023-67-3-183-188}{\it Classical solution of the Cauchy problem for a quasi-linear wave equation with discontinuous initial conditions}, Doklady of the National Academy of Sciences of Belarus, {\bf 67}:3 (2023), 183--188.

\bibitem{cau_nl_3} V.I. Korzyuk, J.V. Rudzko,
\href{https://doi.org/10.1007/s10958-024-07526-x}{\it Classical and mild solutions of the Cauchy problem for a mildly quasilinear wave equation with discontinuous and distributional initial conditions}, J. Math. Sci., {\bf 286}:3 (2024), 535--559.

\bibitem{cau_nla_1} C.D. Sogge,
\href{https://doi.org/10.1080/03605309308820994}{\it On local existence for nonlinear wave equations satisfying variable coefficient null conditions}, Communications in Partial Differential Equations, {\bf 18}:11 (1993), 1795--1821.

\bibitem{cau_ener_1} J. Li, S. Chai,
\href{https://doi.org/10.1016/j.na.2014.08.021}{\it Energy decay for a nonlinear wave equation of variable coefficients with acoustic boundary conditions and a time-varying delay in the boundary feedback}, Nonlinear Analysis, {\bf 112} (2015), 105--117.

\bibitem{cau_ener_2} Y.-X. Liu,
\href{https://doi.org/10.1080/00036811.2020.1849628}{\it Uniform stabilization of a variable coefficient wave equation with nonlinear damping and acoustic boundary}, Applicable Analysis, {\bf 101}:9 (2020), 3347--3364.

\bibitem{cau_ener_3} L. Yan, L. Sun
\href{https://doi.org/10.1002/mma.8554}{\it General stability and exponential growth of nonlinear variable coefficient wave equation with logarithmic source and memory term}, Mathematical Methods in the Applied Sciences, {\bf 46}:1 (2023), 879--894.

\bibitem{cau_nl_4} B.L. Ro\v{z}destvenski\u{\i}, N.N. Janenko
\href{https://bookstore.ams.org/mmono-55}{\it Systems of Quasilinear Equations and Their Applications to Gas Dynamics}, Am. Math. Soc., Providence, 1983.

\bibitem{cau_eik_1} M.I. Belishev, A.L. Pestov,
\href{https://doi.org/10.1007/s10958-010-9948-3}{\it Forward dynamical problem for the Timoshenko beam}, J. Math. Sci., {\bf 167} (2010), 603--621.

\bibitem{cau_eik_2} A.V. Zurov,
\href{https://doi.org/10.1007/s10958-005-0185-0}{\it Effects associated with the coincidence of velocities in a two-velocity dynamical system}, J. Math. Sci., {\bf 127}:6 (2005), 2364--2373.

\bibitem{par1} V.A. Trenogin,
{\it Global invertibility of nonlinear operators and the method of continuation with respect to a parameter}, Dokl. Math., {\bf 54}:2 (1996), 730--732.

\bibitem{par2} V.A. Trenogin,
\href{https://doi.org/10.1007/978-1-4899-1552-8_12}{\it Invertibility of nonlinear operators and parameter continuation method}, In: Ramm A.G. (ed.) \textit{Spectral and Scattering Theory}, Springer, Boston, 1998, 189--197.

\bibitem{sch1} V.I. Korzyuk, J.V. Rudzko
\href{https://doi.org/10.26907/2541-7746.2024.3.377-394}{\it Initial-boundary value problem with Dirichlet and Wentzell conditions for a mildly quasilinear biwave equation}, Uchenye Zapiski Kazanskogo Universiteta. Seriya Fiziko-Matematicheskie Nauki, {\bf 166}:3 (2024), 377--394.

\bibitem{sch2} V.I. Korzyuk, J.V. Rudzko
\href{https://doi.org/10.1134/S199508022460643X}{\it Initial-boundary value problem for a third-order semilinear hyperbolic equation with the wave operator}, Lobachevskii Journal of Mathematics, {\bf 45}:11 (2024), 5569--5580.

\bibitem{sch3} V.I. Korzyuk, J.V. Rudzko
\href{https://doi.org/10.36535/2782-4438-2024-235-40-56}{\it Classical solution of a mixed problem with the Dirichlet and Neumann conditions for a nonlinear biwave equation}, Itogi Nauki i Tekhniki. Sovrem. Mat. Pril. Temat. Obz., {\bf 235} (2024), 40--56.

\bibitem{slim} V.I. Korzyuk, J.V. Rudzko,
\href{https://doi.org/10.3103/S1066369X24700178}{\it Classical solution to the Cauchy problem for a semilinear 
hyperbolic equation in the case of two independent variables}, Russian Mathematics, {\bf 68}:3 (2024), 41--52.

%\bibitem{1} D.G. Northcott,
%\href{https://doi.org/10.1017/CBO9780511565908}{\it Ideal Theory}, Cambridge University Press, Cambridge, 1953.

%\bibitem{2} I.I. Ivanov,
%\href{https://doi.org/10.1002/cpa.20282}{\it Some properties of commutative algebras}, Siberian Electronic
%Mathematical Reports, {\bf 1} (2004), 142--143.

%\bibitem{3} A.V. Tetenov, M. Samuel, D.A. Vauilin,
%\href{https://doi.org/10.21538/0134-4889-2017-23-4-281-291}{\it On dendrites generated by polyhedral systems and their ramification points}, Trudy Inst. Mat. Mekh. UrO RAN, \textbf{23}:4 (2017), 281--291.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}

\thispagestyle{empty}

\title[Classical Solution of the Cauchy Problem]{\large Classical Solution of the Cauchy Problem for a Mildly Quasilinear Wave Equation with a Variable Coefficient}

\author[V.I. Korzyuk]{{\bf V.I. Korzyuk}}
\author[J.V. Rudzko]{{\bf J.V. Rudzko}\protect\orcidicon{0000-0002-1482-9106}}

\address{Viktor Ivanovich Korzyuk  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Belarusian State University,
\newline\hphantom{iii} Nezavisimosti Ave., 4,
\newline\hphantom{iii} 220000, Minsk, Belarus
\newline\hphantom{iii} Institute of Mathematics of the National Academy of Sciences of Belarus,
\newline\hphantom{iii} Surhanava St., 11,
\newline\hphantom{iii} 220000, Minsk, Belarus}%
\email{\textcolor{blue}{korzyuk@bsu.by}}%

\address{Jan Viaczaslavavicz Rudzko  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Institute of Mathematics of the National Academy of Sciences of Belarus,
\newline\hphantom{iii} Surhanava St., 11,
\newline\hphantom{iii} 220000, Minsk, Belarus}%
\email{\textcolor{blue}{janycz@yahoo.com}}%

\thanks{\sc Korzyuk, V.I., Rudzko, I.I.,
Classical Solution of the Cauchy Problem for a~Mildly Quasilinear Wave Equation with a Variable Coefficient}
\thanks{\copyright \ 2025 Korzyuk V.I., Rudzko J.V.}
\thanks{\rm The~authors' research was supported by~the~Moscow Center for~Fundamental and Applied Mathematics of~M.~V.~Lomonosov Moscow State University under agreement no.~075-15-2025-345.}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%

\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc P.P. Petrov}
({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} We consider the Cauchy problem for a mildly quasilinear wave equation with a variable coefficient in the upper half-plane. We use the method of characteristics to construct the solution in an implicit analytical form as a solution of an integro-differential equation. We study the solvability of this equation, as well as the dependence on the initial data and the smoothness of its solutions. We prove the uniqueness of the solution for the problem in question and establish the conditions under which its classical solution exists.\medskip

\noindent{\bf Keywords:} mildly quasilinear wave equation, Cauchy problem, classical solution, apriori estimates, parameter continuation method.
 \end{quote}
}

\bigskip

\section{Statement of the problem}

In the domain $Q=(0, \infty) \times \mathbb R$ of~two independent variables $(t,x)\in \overline{Q}\subset {{\mathbb{R}}^{2}}$, for~the nonlinear equation 
\begin{equation}
\partial_{t}^{2} u(t,x) - a^2(x) \partial _{x}^{2} u(t,x) = f(t,x,u(t,x), \partial_t u(t,x), \partial_x u(t,x)),\ \ (t,x)\in Q, \label{korzyuk_eq_base}
\end{equation}
we consider the~following Cauchy problem with the~initial conditions
\begin{equation}
u(0, x) = \varphi(x), \quad {{\partial }_{t}}u(0,x)=\psi(x),\quad x\in \mathbb{R}, \label{korzyuk_cau_base}
\end{equation}
where $0 < a_\mathrm{min} \leq a(x) \leq a_\mathrm{max}$ for all $x \in \mathbb{R}$, $f$ is a function given on the~set $\overline{Q} \times\mathbb{R}^3$, and $\varphi$ and $\psi$ are a real-valued functions defined on the set $\mathbb{R}$. %We also assume
%\begin{equation}
%     \int\limits_0^{\pm \infty} \frac{ds}{a(s)} = \pm \infty. \label{korzyuk_cau_add_cond}
%\end{equation}

Equations of form~\eqref{korzyuk_eq_base} are found in mechanics (nonlinear vibrations of strings and rods)~\cite{mech, mech2} and thermodynamics (the Maxwell--Cattaneo model of heat conduction)~\cite{thermo}.

Work~\cite{cau_0} obtained in an explicit analytical form a classical solution to the problem
\begin{align*}
     &(\partial_t - a(t,x) \partial_x - \partial_t (\ln a(t,x))) (\partial_t - a(t,x) \partial_x) u(t,x) = \\
     &\quad = a(t,x)\partial_x a(t,x) \partial_x u(t,x) - \partial_t a(t,x) \partial_x u(t,x) + \partial_t (\ln a(t,x)) \times \\
     &\quad \times (\partial_t u(t,x) + a(t,x) \partial_x u(t,x)) - b(t,x) \partial_t u(t,x) - c(t,x) \partial_x u(t,x) \, - \\
     &\quad  - q(t,x) u(t,x), \quad (t,x) \in Q,\\
     &u(0,x) = \varphi(x), \quad {{\partial }_{t}}u(0,x)=\psi(x),\quad x\in \mathbb{R}.
\end{align*}
Paper~\cite{cau_1} considered the following general linear case of problem~(1)--(2)
\begin{align*}
     &\partial_{t}^{2} u(t,x) - a^2(t, x) \partial _{x}^{2} u(t,x) + a^{(1)}(t,x) \partial_t u(t,x) + a^{(2)}(t,x) \partial_x u(t,x) \, + \\ &\quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad \quad + a^{(3)}(t,x) u(t,x) = f(t,x), \\
     &u(s(x), x) = \varphi(x), \quad {{\partial }_{t}} u(s(x), x)=\psi(x),
\end{align*}
where the coefficient $a(t, x)$ does not vanish at any point. Communication~\cite{cau_2} studied the problem
\begin{align*}
     &\partial_{t}^{2} u(t,\mathbf x) - \sum\limits_{i=1}^n \sum\limits_{j=1}^n a_{i,j} \partial_{x_i}\partial_{x_j} u(t, \mathbf x)  = 0,\quad (t,\mathbf x) \in (0,\infty) \times \mathbb R^n, \\
     &u(0, \mathbf x) = \varphi(\mathbf x), \quad {{\partial }_{t}} u(0, \mathbf x)=\psi(\mathbf x),\quad \mathbf x \in \mathbb R^n,
\end{align*}
when the initial data $\varphi$ and $\psi$ belong to the Schwartz space $\mathcal{S}(\mathbb R^n)$. Article~\cite{cau_3} considered problem~(1)--(2) with $f \equiv 0$ in the Triebel--Lizorkin space.

The observability issues for the linear equations of form~(1) were considered in works~\cite{cau_4, cau_5}. Strichartz type estimates for linear equations of form~\eqref{korzyuk_eq_base} were obtained in paper~\cite{cau_6}. Weighted $L^2$-estimates for solutions of the equation
$$
\partial_t^2 u(t, \mathbf x) - \Delta u(t, \mathbf x) + c(\mathbf x) \partial_t u(t, \mathbf x) = 0
$$
were derived in~\cite{cau_7}.

Communication~\cite{cau_nl_1} studied problem~\eqref{korzyuk_eq_base}--\eqref{korzyuk_cau_base} in the case $a(x) = \mathrm{const}$. Works~\cite{cau_nl_2, cau_nl_3} investigated problem~(1)--(2) with discontinuous initial data in the case $a(x) = \mathrm{const}$. 

Local existence and uniqueness of a weak solution of the Cauchy problem for the hyperbolic equation
\begin{multline*}
\bigg(\sum\limits_{0 \leq j, k \leq n} g^{(j,k)}(\mathbf x) \partial_{x_j} \partial_{x_k} + \sum\limits_{j=0}^n b^{(j)}(\mathbf x) \partial_{x_j} \partial_{x_j} + c(\mathbf x)\bigg) u(\mathbf x) = \\
= F(u(\mathbf x), \partial_{x_0} u(\mathbf x), \partial_{x_1} u(\mathbf x), \dots, \partial_{x_n} u(\mathbf x))
\end{multline*}
was established in~\cite{cau_nla_1}. Energy estimates for solutions to mixed problem of the nonlinear equations of the form
$$
\partial_{t t} u(t,\mathbf x) + \mathrm{div} (a(\mathbf x) \nabla u(t,\mathbf x)) + \cdots = 0
$$
were derived in~\cite{cau_ener_1, cau_ener_2, cau_ener_3}.

Note that Eq.~\eqref{korzyuk_eq_base} can be reduced to a first-order semilinear hyperbolic system
\begin{align}
     &\partial_t u(t,x) = q(t,x), \notag \\
     &\partial_t p(t,x) - \partial_x q(t,x) = 0,  \label{korzyuk_syst_base} \\
     &\partial_t q(t,x) - a^2(x) \partial_x p(t,x) = f(t,x,u(t,x), q(t,x), p(t,x)), \notag
\end{align}
with respect to the unknown functions $u$, $q = \partial_t u$, and $p = \partial_x u$. Results on the existence and uniqueness of solutions in the broad sense solutions to the Cauchy problem for system~\eqref{korzyuk_syst_base} are well known and presented in book~\cite{cau_nl_4}. However, generally speaking, this book only establishes local solutions. They will be global classical provided that they are bounded in any characteristic triangle and the given functions $f$, $\varphi$, $\varphi'$, and $\psi$ are continuously differentiable. The answer to the question ``What conditions must be imposed on these functions to obtain global classical solutions?'' is not available in~\cite{cau_nl_4}.

\section{Linear problem} 

We begin by considering the following simplest linear case of problem ~\eqref{korzyuk_eq_base}--\eqref{korzyuk_cau_base} 
\begin{equation}\label{korzyuk_lin}
     f(t,x,u,u_t,u_x) = g(t,x).
\end{equation}

Following~\cite{cau_eik_1, cau_eik_2}, we denote the eikonal
\begin{equation}\label{korzyuk_eik}
     T(x) := \int\limits_0^x \frac{ds}{a(s)}.
\end{equation}
Let $X$ be the inverse of the function $T$. The function $X\colon \mathbb R \mapsto \mathbb R$ exists due to the condition $0 < a_\mathrm{min} \leq a(x) \leq a_\mathrm{max}$, which implies that the function $T$ is strictly increasing and the integrals
\begin{equation}\label{korzyuk_cau_add_cond_iii}
     \int\limits_0^{\pm \infty} \frac{ds}{a(s)}
\end{equation}
diverge, which means that the function $T$ biectivly maps the set $\mathbb{R}$ into itself.

Eikonal~\eqref{korzyuk_eik} defines the characteristics of Eq.~\eqref{korzyuk_eq_base} as follows: $t \pm T(x) = \mathrm{const}$. The function $X$ has the representation
$$
X(x) = \int\limits^x_0 a(X(\xi)) \, d\xi.
$$
Direct differentiation yields the following formulas:
\begin{gather}
     T'(x) = \frac{1}{a(x)}, \quad T''(x) = -\frac{a'(x)}{a^2(x)}, \label{korzyuk_eik_dd} \\ 
     X'(x) = a(X(x)), \quad X''(x) = a'(X(x)) a(X(x)). \label{korzyuk_eik_dd2}
\end{gather}
Thus, if $a \in C^1(\mathbb R)$, then $T \in C^2(\mathbb R)$ and $X \in C^2(\mathbb R)$.

We will look for a solution $u$ of problem~\eqref{korzyuk_eq_base}, \eqref{korzyuk_cau_base}, and~\eqref{korzyuk_lin} having the form
\begin{equation}\label{korzyuk_anz}
    u(t,x) = \sqrt{\frac{a(x)}{a(0)}} v(t,T(x)), \quad (t,x) \in \overline Q.
\end{equation}
Taking into account~\eqref{korzyuk_eik_dd} and~\eqref{korzyuk_eik_dd2}, we substitute~\eqref{korzyuk_anz} into Eq.~\eqref{korzyuk_cau_base} and get
\begingroup
\allowdisplaybreaks
\begin{multline}\label{korzyuk_new_eq}
     \partial_t^2 v(t, x) - \partial_x^2 v(t, x) = \sqrt{\frac{a(0)}{a(X(x))}} g(t,X(x)) \, - \\ -\frac{1}{4} v(t,x) \left(a'(X(x))^2-2 a(X(x)) a''(X(x))\right)\! = \sqrt{\frac{a(0)}{a(X(x))}} g(t,X(x)) \, - \\ -\frac{1}{4} \sqrt{\frac{a(0)}{a(X(x))}} u(t,X(x)) \left(a'(X(x))^2-2 a(X(x)) a''(X(x))\right)\!, \:\: (t,x) \in Q.
\end{multline}
\endgroup
The initial conditions for the function $v$ are
\begin{align}
     v(0,x) &= \sqrt{\frac{a(0)}{a(X(x))}} \varphi(X(x)), \:\: x\in\mathbb R, \label{korzyuk_nc1} \\
     \partial_t v(0,x) &= \sqrt{\frac{a(0)}{a(X(x))}} \psi(X(x)), \:\: x\in\mathbb R. \label{korzyuk_nc2} 
\end{align}

We apply~\cite[Lemma~1]{cau_nl_1} to problem~\eqref{korzyuk_new_eq}--\eqref{korzyuk_nc2} to obtain
\begingroup
\allowdisplaybreaks
\begin{multline}
     u(t,x) = \frac{\sqrt{a(x)} }{2} \left(\frac{\varphi (X(T(x)-t))}{\sqrt{a(X(T(x)-t))}}+\frac{\varphi
     (X(T(x) + t))}{\sqrt{a(X(T(x) + t))}}\right) + \\
     + \frac{\sqrt{a(x)}}{2} \int\limits_{T(x)-t}^{T(x)+t} \frac{\psi (X(\xi ))}{\sqrt{a(X(\xi ))}} \, d\xi 
     + \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{T(x)-t+\tau}^{T(x)+t-\tau} \Bigg[ \frac{g(\tau ,X(\xi ))}{\sqrt{a(X(\xi))}} + \\
     + \frac{u(\tau ,X(\xi )) \left(2 a(X(\xi )) a''(X(\xi ))-a'(X(\xi ))^2\right)}{4 \sqrt{a(X(\xi ))}} \Bigg] d\xi, \quad (t,x) \in \overline Q. \label{korzyuk_ieq_pre}
\end{multline}
\endgroup

Thus, we have proved the following assertion.

\begin{assertion}\label{korzyuk_assert_1}
     Let the conditions $a \in C^2(\mathbb R)$, $g \in C^1(\overline Q)$, $\varphi \in C^2(\mathbb R)$, and $\psi \in C^1(\mathbb R)$ be satisfied. Then a solution $u \in C^2(\overline Q)$ to problem~\eqref{korzyuk_eq_base}, \eqref{korzyuk_cau_base}, and~\eqref{korzyuk_lin} satisfies Eq. (5) in $\overline Q$.
\end{assertion}

\section{Integro-differential equation} 

We consider the integral equation
\begingroup
\allowdisplaybreaks
\begin{multline}
     u(t,x) = K[u](t,x) = \frac{\sqrt{a(x)} }{2} \times \\ \times \left(\frac{\varphi (X(T(x)-t))}{\sqrt{a(X(T(x)-t))}}+\frac{\varphi
     (X(T(x) + t))}{\sqrt{a(X(T(x) + t))}}\right) + \\
     + \frac{\sqrt{a(x)}}{2} \int\limits_{T(x)-t}^{T(x)+t} \frac{\psi (X(\xi ))}{\sqrt{a(X(\xi ))}} \, d\xi 
     + \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{T(x)-t+\tau}^{T(x)+t-\tau} \Psi(\tau ,X(\xi ),\\
     u(\tau ,X(\xi )), \partial_t u(\tau ,X(\xi )), \partial_x u(\tau ,X(\xi ))) \, d\xi, \quad (t,x) \in \overline Q. \label{korzyuk_ieq}
\end{multline}
\endgroup
where
$$
\Psi(t,x,u,u_t,u_x) = \frac{f(t,x,u,u_t,u_x)}{\sqrt{a(x)}} + \frac{u \left(2 a(x) a''(x)-a'(x)^2\right)}{4 \sqrt{a(x)}}.
$$

The following theorem on the equivalence of the ``differential'' and ``integral'' formulations of problem~\eqref{korzyuk_eq_base}--\eqref{korzyuk_cau_base} holds.

\begin{theorem}\label{korzyuk_th_1}
 Let the conditions $a \in C^2(\mathbb R)$, $f \in C^2(\overline Q \times \mathbb R)$, $\varphi \in C^2(\mathbb R)$, and $\psi \in C^1(\mathbb R)$ be satisfied. The function $u$ belongs to the class $C^2(\overline Q)$ and satisfies Eq.~\eqref{korzyuk_eq_base} in $\overline Q$ and conditions~\eqref{korzyuk_cau_base} on the set $\mathbb{R}$ if and only if it is a continuously differentiable solution of Eq.~\eqref{korzyuk_ieq} in $\overline Q$. 
\end{theorem}

\begin{proof}
     1. Let the function $u$ belong to the class $C^2(\overline Q)$ and solve Eq.~\eqref{korzyuk_eq_base} with conditions~\eqref{korzyuk_cau_base}. We seek a representation of the solution $u$ of Eq.~(1) problem~\eqref{korzyuk_eq_base}--\eqref{korzyuk_cau_base} in the form $u = v + w$, where $v$ is a solution to the problem
     \begingroup
\allowdisplaybreaks
     \begin{align*}
          &\partial _{t}^{2} u(t,x) - a^2(x) \partial _{x}^{2} u(t,x) = 0, \quad (t,x)\in Q, \\
          &u(0,x) = \varphi(x), \quad {{\partial }_{t}} u(0,x)=\psi(x), \quad x\in \mathbb{R},
     \end{align*}
     \endgroup
     and $w$ is a solution to the equation
     \begingroup
\allowdisplaybreaks
     \begin{multline*}
          \partial _{t}^{2} w(t,x) - a^2(x) \partial _{x}^{2} w(t,x) = \\ = f(t,x,(v + w)(t,x), \partial_t (v + w)(t,x), \partial_x (v + w)(t,x)),\quad (t,x)\in Q,
     \end{multline*}
     \endgroup
     with conditions
     $$
     w(0,x) = {{\partial }_{t}} w(0,x)=0, \quad x\in \mathbb{R}.
     $$
     Assertion~\ref{korzyuk_assert_1} allows us to write Eq.~\eqref{korzyuk_ieq}.

     2. Assume that representation~\eqref{korzyuk_ieq} hold for the function $u \in C^1(\overline Q)$. Then, by virtue of the smoothness conditions $a \in C^2(\mathbb R)$, $f \in C^2(\overline Q \times \mathbb R)$, $\varphi \in C^2(\mathbb R)$, and $\psi \in C^1(\mathbb R)$, similar to~\cite{cau_1}, it follows that the function $u$ belongs to the class $C^2(\overline Q)$. We substitute representation~\eqref{korzyuk_ieq} into Eq.~\eqref{korzyuk_eq_base} and conditions~\eqref{korzyuk_cau_base} and verify that the function $u$ satisfies this equation in $\overline Q$ and these conditions in $\mathbb R$.
\end{proof}

Before solving Eq.~\eqref{korzyuk_ieq}, we will study some properties of the operator $K$. By making the nondegenerate change of variables $X(\xi) = s$, we can rewrite
\begingroup
\allowdisplaybreaks
\begin{multline}
 K[u](t,x) = \frac{\sqrt{a(x)} }{2} \times \\ \times \left(\frac{\varphi (X(T(x)-t))}{\sqrt{a(X(T(x)-t))}}+\frac{\varphi
 (X(T(x) + t))}{\sqrt{a(X(T(x) + t))}}\right) + \\
 + \frac{\sqrt{a(x)}}{2} \int\limits_{T(x)-t}^{T(x)+t} \frac{\psi (X(\xi ))}{\sqrt{a(X(\xi ))}} \, d\xi 
 + \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \Phi(\tau , s,\\
 u(\tau ,s), \partial_t u(\tau ,s), \partial_x u(\tau ,s)) \, ds, \quad (t,x) \in \overline Q. \label{korzyuk_ieq_op_def_rw}
\end{multline}
\endgroup
where 
$$
\Phi(t,x,u,u_t,u_x) = \frac{\Psi(t,x,u,u_t,u_x)}{a(x)}.
$$
Now, we see that the operator $K$ maps the space $C^1(\Omega_{(t_0,x_0)})$ into the space $C^1(\Omega_{(t_0,x_0)})$, where
\begin{equation*}
     \Omega_{(t_0,x_0)} = \{(t, x) : 0 \leq t \leq t_0 \land X(T(x_0)-t_0+t) \leq x \leq X(T(x_0)-t_0+t)\}
\end{equation*}
Note that the set $\Omega_{(t_0,x_0)}$ is the characteristic triangle of Eq.~\eqref{korzyuk_eq_base}.

At the same time $\Omega_{(t_0,x_0)} \subseteq \Omega_{(t_1,x_1)}$ if $(t_0,x_0) \in \Omega_{(t_1,x_1)}$. It follows from the fact that the representation
$$
\Omega_{(t_0,x_0)} = \bigcup\limits_{t \in [0, t_0]} \{(t, x) : T(x_0)-t_0+t \leq X(x) \leq T(x_0)+t_0-t \}
$$
holds and $T$ is the strictly increasing function. Thus, since $(n,0) \in \Omega_{(n+1,0)}$, $n \in \mathbb{N}$, we have $\Omega_{(n,0)} \subset \Omega_{(n+1,0)}$. It implies $\Omega_{(1,0)} \subset \Omega_{(2,0)} \subset \Omega_{(3,0)} \subset \cdots$. Moreover, we have
\begin{equation}\label{korzyuk_izch}
     \bigcup\limits_{i=1}^\infty \Omega_{(i,0)} = \overline Q.
\end{equation}

From the definition of the set $\Omega_{(t_0,x_0)}$, we deduce
$$
\Omega_{(t_0,x_0)} \subset \mathcal{B}_{(t_0,x_0)} := [0, t_0] \times [X(T(x_0) - t_0), X(T(x_0) + t_0)].
$$ 

Now, we will solve Eq.~\eqref{korzyuk_ieq} using the parameter continuation method~\cite{par1, par2} according to the scheme set forth in~\cite{sch1, sch2, sch3}. We rewrite Eq.~\eqref{korzyuk_ieq} in the operator form
\begin{equation}\label{korzyuk_ieq_op}
 u(t,x) = J[u](t,x) + h(t, x),
\end{equation}
where
\begingroup
\allowdisplaybreaks
\begin{align*}
 h(t,x) &= \frac{\sqrt{a(x)}}{2} \left(\frac{\varphi (X(T(x)-t))}{\sqrt{a(X(T(x)-t))}} +\frac{\varphi
 (X(T(x) + t))}{\sqrt{a(X(T(x) + t))}}\right) + \\
     & + \frac{\sqrt{a(x)}}{2} \int\limits_{T(x)-t}^{T(x)+t} \frac{\psi (X(\xi ))}{\sqrt{a(X(\xi ))}} \, d\xi, \\
 J[u](t, x) &= \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \Phi(\tau ,\xi ,\\
     &u(\tau ,\xi), \partial_t u(\tau ,\xi), \partial_x u(\tau ,\xi)) \, d\xi, \quad (t,x) \in \overline Q. 
\end{align*}
\endgroup
For Eq.~\eqref{korzyuk_ieq_op}, we consider a family of equations with the parameter $\varepsilon \in [0,1]$,
\begin{equation}\label{korzyuk_ieq_op_e}
     u_\varepsilon(t, x) - \varepsilon(J[u_\varepsilon] - J[0])(t,x) = h(t,x) + J[0](t,x), \quad (t,x) \in \overline Q. 
\end{equation}
It is easy to see that any solution $u_\varepsilon$ to Eq.~\eqref{korzyuk_ieq_op_e} with $\varepsilon = 1$ is a solution to Eq.~\eqref{korzyuk_ieq_op} and vice versa. Our goal is to solve Eq.~\eqref{korzyuk_ieq_op_e} with $\varepsilon = 1$. To do this, we will use the parameter continuation method. First, we must show that the operator on the left-hand side of Eq.~\eqref{korzyuk_ieq_op_e} satisfies the Lipschitz condition with some constant. Second, we must obtain an a priori estimate of a certain type.

Assume that the function $f$ satisfies the Lipschitz condition 
\begin{equation}\label{korzyuk_lip_cond_f}
|f(t,x,u_1,u_2,u_3) - f(t,x,z_1,z_2,z_3)| \leq L(t,x) \sum\limits_{i=1}^3 |u_i - z_i|,
\end{equation}
where $L \in C(\overline Q)$. Then, the function $\Phi$ also satisfies the Lipschitz condition 
\begin{equation}\label{korzyuk_lip_cond_fff}
|\Phi(t,x,u_1,u_2,u_3) - \Phi(t,x,z_1,z_2,z_3)| \leq L_\Phi(t,x) \sum\limits_{i=1}^3 |u_i - z_i|,
\end{equation}
where 
$$
L_\Phi(t,x) = \frac{L(t, x)}{a^{3/2}(x)} + \left|\frac{2 a(x) a''(x)-a'(x)^2}{4 a^{3/2}(x)}\right|\!.
$$

Now, let's estimate the difference $|J[u_1](t,x) - J[u_1](t,x)|$. We have
\begingroup
\allowdisplaybreaks
\begin{multline}\label{korzyuk_est_1}
     |J[u_1](t,x) - J[u_1](t,x)| = \Bigg | \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \Phi(\tau ,\xi,u_1(\tau ,\xi ), \\
     \partial_t u_1(\tau ,\xi), \partial_x u_1(\tau ,\xi)) \, d\xi - \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \Phi(\tau ,\xi,\\
     u_2(\tau ,\xi), \partial_t u_2(\tau ,\xi), \partial_x u_2(\tau ,\xi)) \, d\xi \Bigg| \leq \frac{\sqrt{a(x)}}{2} \times \\
     \times \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \big|\Phi(\tau ,\xi,u_1(\tau ,\xi), \partial_t u_1(\tau ,\xi), \partial_x u_1(\tau ,\xi)) - \\
     - \Phi(\tau ,\xi,u_2(\tau ,\xi), \partial_t u_2(\tau ,\xi), \partial_x u_2(\tau ,\xi)) \big| d\xi \leq \\
     \leq \frac{\sqrt{a(x)}}{2} \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau)}^{X(T(x)+t-\tau)} \big[ L_\Phi(\tau ,\xi) |U(\tau ,\xi)| + L_\Phi(\tau ,\xi) \times \\ \times |\partial_t U(\tau ,\xi)|
     + L_\Phi(\tau ,\xi) |\partial_x U(\tau ,\xi)|\big] d\xi,
\end{multline}
\endgroup
where $U = u_1 - u_2$. Estimate~\eqref{korzyuk_est_1} imply
\begingroup
\allowdisplaybreaks
\begin{multline}\label{korzyuk_est_2}
     |J[u_1](t,x) - J[u_1](t,x)| \leq \frac{\sqrt{\| a \|_{C(\Omega_m)}} \| L_\Phi \|_{C(\Omega_m)} \| u_1 - u_2 \|_{C^1(\Omega_m)} \mu(\Omega_m)}{2}, \\ (t,x) \in \Omega_m, \quad m \in \mathbb{N},
\end{multline} 
\endgroup   
where $\Omega_m = \Omega_{(m, 0)}$ and $\mu$ is the standard Lebesgue measure. 

Similarly, we derive the following estimates:
\begingroup
\allowdisplaybreaks
\begin{align}
     &\| \partial_t J[u_1] - \partial_t J [u_1] \|_{C(\Omega_m)} \leq C^{(1)} \| u_1 - u_2 \|_{C^1(\Omega_m)}, \label{korzyuk_est_t} \\ 
     &\| \partial_x J[u_1] - \partial_x J [u_1] \|_{C(\Omega_m)} \leq C^{(2)} \| u_1 - u_2 \|_{C^1(\Omega_m)}, \label{korzyuk_est_x}
\end{align}
\endgroup
where $C^{(1)}$ and $C^{(2)}$ are constants depending only on the functions $a$ and $L_\Phi$. In this case, one has the inequality
$$
\| J[u_1] - J [u_1] \|_{C^1(\Omega_m)} \leq \mathfrak{L} \| u_1 - u_2 \|_{C^1(\Omega_m)}
$$
where 
$$
\mathfrak{L} = \frac{\sqrt{\| a \|_{C(\Omega_m)}} \| L_\Phi \|_{C(\Omega_m)} \mu(\Omega_m)}{2} + C^{(1)} + C^{(2)},
$$
It shows that the operator $J \colon C^1(\Omega_m) \mapsto C^1(\Omega_m)$ satisfies the Lipschitz condition with the constant $\mathfrak L$.

The task now is to obtain an a priori estimate of the following form:
\begin{equation}\label{korzyuk_apr_f}
     \| u_\varepsilon\|_{C^1(\Omega_m)} \leq C \| h + J[0] \|_{C^1(\Omega_m)}
\end{equation}
for all possible solutions $u_\varepsilon$ of Eq.~\eqref{korzyuk_ieq_op_e} for any $\varepsilon \in [0,1]$, where $C$ is some constant independent of $h$, $u_\varepsilon$, and $\varepsilon$.
%h(t,x) + J[0](t,x) - \varepsilon(J[u_\varepsilon] - J[0])(t,x)

For a fixed $m \in \mathbb{N}$, let us introduce a vector function
\begingroup
\allowdisplaybreaks
\begin{multline*}
\mathbf{v}^{(m)}:\left[X(-m), X(m)\right] \ni t \mapsto \mathbf{v}^{(m)}(t) = \\
=(x \mapsto (R_{m}[\Delta u](t, x), R_{m}[\partial_{t} \Delta u](t, x), R_{m}[\partial_{x} \Delta u](t, x))) \in \\
\in PC\left(\left[X(-m), X(m)\right]\right) \times PC\left(\left[X(-m), X(m)\right]\right) \times PC\left(\left[X(-m), X(m)\right]\right)
\end{multline*}
\endgroup
where $R_{m}[h]$ is the extension by zero of the restriction $h|_{\Omega_{m}}$ of the function $h$, and $PC(\Omega)$ is a space of piecewise continuous functions on $\Omega$.

Thus, we have 
\begingroup
\allowdisplaybreaks
\begin{multline}\label{korzyuk_ue_1}
     |u_\varepsilon(t,x)| \leq |h(t,x) + J[0](t,x) - \varepsilon(J[u_\varepsilon] - J[0])(t,x)| \leq \| h + J[0] \|_{C(\Omega_{m})} \, + \\
     + |(J[u_\varepsilon] - J[0])(t,x)| \leq \| h + J[0] \|_{C(\Omega_{m})} + \frac{\sqrt{a(x)}}{2} \times \\
     \times \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau}^{X(T(x)+t-\tau)} \big|\Phi(\tau ,\xi,u_\varepsilon(\tau ,\xi), \partial_t u_\varepsilon(\tau ,\xi), \partial_x u_\varepsilon(\tau ,\xi)) \,- \\
     - \Phi(\tau ,\xi,0, 0, 0) \big| d\xi \leq \| h + J[0] \|_{C(\Omega_{m})} + \frac{\sqrt{a(x)}}{2} \times \\
     \times \int\limits_0^t d\tau \int\limits_{X(T(x)-t+\tau}^{X(T(x)+t-\tau)} L_\Phi(\tau ,\xi) (u_\varepsilon(\tau ,\xi) + \partial_t u_\varepsilon(\tau ,\xi) \, + \\
     + \partial_x u_\varepsilon(\tau ,\xi)) d\xi \leq \| h + J[0] \|_{C(\Omega_{m})} + \sqrt{\| a\|_{C(\Omega_{m})}} \| L_\Phi\|_{C(\Omega_{m})} \times \\
     \times \int\limits_0^t \max _{0 \leq \xi \leq \tau}\|\mathbf{v}^{(m)}(\xi)\| \, d\tau, \quad (t,x) \in \Omega_{m}. 
\end{multline}
\endgroup
Estimate~\eqref{korzyuk_ue_1} yields
\begin{equation}\label{korzyuk_vvv_e1} 
     \|\mathbf{v}_{1}^{(m)}(t)\| \leq \| h + J[0] \|_{C(\Omega_{m})} + C_v^{(1, m)} \int\limits_0^t \max _{0 \leq \xi \leq \tau}\|\mathbf{v}^{(m)}(\xi)\| \, d\tau, \quad t \in [0, m].
\end{equation}
where $C_v^{(1, m)} = \| L_\Phi\|_{C(\Omega_{m})} \sqrt{\| a\|_{C(\Omega_{m})}}$. In a similar way, we derive the estimates
\begingroup
\allowdisplaybreaks
\begin{align}
     \|\mathbf{v}_{2}^{(m)}(t)\| \leq \| \partial_t(h + J[0]) \|_{C(\Omega_{m})} + C_v^{(2, m)} \int\limits_0^t \max _{0 \leq \xi \leq \tau}\|\mathbf{v}^{(m)}&(\xi)\| \, d\tau, \notag \\ &t \in [0, m], \label{korzyuk_vvv_e2} \\
     \|\mathbf{v}_{3}^{(m)}(t)\| \leq \| \partial_x(h + J[0]) \|_{C(\Omega_{m})} + C_v^{(3, m)} \int\limits_0^t \max _{0 \leq \xi \leq \tau}\|\mathbf{v}^{(m)}&(\xi)\| \, d\tau, \notag \\ &t \in [0, m], \label{korzyuk_vvv_e3}
\end{align}
\endgroup

Estimates~\eqref{korzyuk_vvv_e1}--\eqref{korzyuk_vvv_e3} imply
\begin{equation*}
     \|\mathbf{v}^{(m)}(t)\| \leq \| h + J[0] \|_{C^1(\Omega_{m})} + C_v^{(m)} \int\limits_0^t \max _{0 \leq \xi \leq \tau}\|\mathbf{v}^{(m)}(\xi)\| \, d\tau, \quad t \in [0, m],
\end{equation*}
where $C_v^{(m)} = C_v^{(1, m)} + C_v^{(2, m)} + C_v^{(3, m)}$. Applying~\cite[p. 39, Lemma 1]{cau_nl_4} to the previous inequality, we obtain
$$
\|\mathbf{v}^{(m)}(t)\| \leq \| h + J[0] \|_{C^1(\Omega_{m})} \exp \!\left(C_v^{(m)} m\right)\!, \quad t \in [0, m].
$$
The last inequality yields estimate~\eqref{korzyuk_apr_f} with $C = \exp \!\left(C_v^{(m)} m\right)$.

Consider the operator $A_\varepsilon$, which acts according to the formula
$$
A_\varepsilon[u] = u(t,x) - \varepsilon(J[u] - J[0]).
$$
Note that the operator $A_\varepsilon$ is Lipschitz-continuous (this follows from the fact that the operator $J$ is Lipschitz-continuous) and coercive in the sense of the paper~\cite{par1} (the coercivity follows from the apriori estimate~\eqref{korzyuk_apr_f}). Furthermore, the function $[0,1] \ni \varepsilon \mapsto A_\varepsilon$ is continuous in the seminorm of the space of Lipschitz-continuous operators~\eqref{korzyuk_apr_f}. Since the operator $A_\varepsilon$ with $\varepsilon = 0$ is continuously invertible in the space $C^1(\Omega_{m})$, then by virtue of~\cite[Theorem 4]{par1}, it is invertible for all $\varepsilon \in [0,1]$ in the same space. Thus, Eq.~\eqref{korzyuk_ieq_op_e} is solvable for $\varepsilon = 1$ on the set $\Omega_{m}$ in the class of continuously differentiable functions. According to the construction, the solution is unique. It means that the equation~\eqref{korzyuk_ieq_op} also has a unique solution in the space $C^1(\Omega_{m})$ for any $h \in C^1(\Omega_{m})$.

Due to~\eqref{korzyuk_izch}, a solution $u^{(\infty)}$ to Eq.~\eqref{korzyuk_ieq_op} in the space $C^1(\overline Q)$ can be constructed in the form $u^{(\infty)} = \lim\limits_{m\to \infty} u^{(m)}$, where $u^{(m)}$ is its solution on the set $\Omega_{m}$. As in~\cite{slim}, one can verify that the function $u^{(\infty)} \colon \overline Q \mapsto \mathbb R$ is indeed the only solution in the class $C^1(\overline Q)$ to Eq.~\eqref{korzyuk_ieq_op}, which continuously depends on the right-hand side.

Thus, we have proven the following theorem.

\begin{theorem}\label{korzyuk_th_2}
     Let the conditions $a \in C^2(\mathbb R)$, $f \in C^1(\overline Q \times \mathbb R)$, $\varphi \in C^1(\mathbb R)$, and $\psi \in C(\mathbb R)$ be satisfied, and let a function $f$ satisfy the Lipschitz condition~\eqref{korzyuk_lip_cond_f}. Then a continuously differentiable solution to Eq.~\eqref{korzyuk_ieq} exists, is unique, and depends continuously on the initial data.
\end{theorem}

\section{Global classical solution} 

\begin{theorem}\label{korzyuk_th_3}
     Let the conditions $a \in C^2(\mathbb R)$, $f \in C^2(\overline Q \times \mathbb R)$, $\varphi \in C^2(\mathbb R)$, and $\psi \in C^1(\mathbb R)$ be satisfied, and let a function $f$ satisfy the Lipschitz condition~\eqref{korzyuk_lip_cond_f}. Problem~\eqref{korzyuk_eq_base}--\eqref{korzyuk_cau_base} has a unique solution defined by formula~\eqref{korzyuk_ieq} in the class $C^2(\overline Q)$.
\end{theorem}

\begin{proof}
It follows from Theorems~\ref{korzyuk_th_1} and~\ref{korzyuk_th_2}.
\end{proof}

\section{Boundedness condition of the variable coefficient} 

In the present paper, we have considered Eq.~\eqref{korzyuk_eq_base} provided 
\begin{equation}\label{korzyuk_ineq_a_max}
 a(x) \leq a_\mathrm{max}
\end{equation}
for all $a \in \mathbb R$. However, this condition can be replaced with the following
\begin{equation}
     \int\limits_0^{\pm \infty} \frac{ds}{a(s)} = \pm \infty. \label{korzyuk_cau_add_cond}
\end{equation}
Let us prove it. We used condition~\eqref{korzyuk_ineq_a_max} only to show that integrals~\eqref{korzyuk_cau_add_cond_iii} diverge when proving the existence of the function $X$. Condition~\eqref{korzyuk_cau_add_cond} means that integrals~\eqref{korzyuk_cau_add_cond_iii} diverge. 

\section{Conclusions} 

In this paper, we derive the necessary and sufficient conditions under which there exists a unique classical solution of the Cauchy problem in the upper half-plane for a mildly quasilinear wave equation with a variable coefficient. We also established the dependence of the smoothness of the solution on the smoothness of the initial data. In the future, we plan to use the results obtained in this research to consider mixed problems for  Eq.~\eqref{korzyuk_eq_base}.

\bigskip

\end{document}
